ACE Weekly Report: 12/29/99





All ACE spacecraft subsystems are performing nominally. A successful attitude maneuver was completed on 12/28/99.  Upcoming attitude maneuvers are scheduled for 01/06/00 and 01/14/00.





On day 357 (12/23/99) at approx. 2300z the UPS supporting the ACE MOR experienced a glitch which caused the MOR system to power cycle.  MAR system was not impacted.   SOHO personnel contacted ACE of this occurrence during ACE off hours. �
�
FOT came in after being contacted to check string status.  The MOR FEP and Workstation 2 were impacted by glitch.  TSM’s were notified and FEP and WS2 were brought back online.  There was no impact to day 358 support.�
�



On 12/24/99 prior to AOS, D27 informed MOC that the Antenna Brake was frozen.  Antenna began tracking about 5 minutes after the scheduled AOS and telemetry was online to MOC 6 min 39 seconds past scheduled BOT. Anomaly report AR G99-0065 associated with  DR# G05617 was generated as a result of the late AOS.





On 12/28/99 station D54 had a delayed AOS due to link controller setup problems - No impact. Executed OCR SC-012 to load TT bin to poke C&DH’s in case of communication problems for Y2K contingency. The poke commands will execute at 00-004-17:00:00 on active and inactive C&DH's which will provide another 96 hours until a reset without ground contact.








